NBER WORKING PAPER SERIES

DETERMINANTS OF VERTICAL INTEGRATION: FINANCE, CONTRACTS, AND REGULATION

Daron Acemoglu Simon Johnson Todd Mitton

Working Paper 11424 http://www.nber.org/papers/w11424

NATIONAL BUREAU OF ECONOMIC RESEARCH 1050 Massachusetts Avenue Cambridge, MA 02138 June 2005

Corresponding author: Simon Johnson, Research Department, 10-700, IMF, 700 19th Street, N.W., Washington, DC 20431; (202)623-8720 phone, (202)589-8720 fax, sjohnson@imf.org. The views expressed herein are those of the author(s) and do not necessarily reflect the views of the National Bureau of Economic Research.

©2005 by Daron Acemoglu, Simon Johnson, and Todd Mitton. All rights reserved. Short sections of text, not to exceed two paragraphs, may be quoted without explicit permission provided that full credit, including © notice, is given to the source.

Determinants of Vertical Integration: Finance, Contracts, and Regulation Daron Acemoglu, Simon Johnson, and Todd Mitton NBER Working Paper No. 11424 June 2005 JEL No. G30, G34, L22, L23

ABSTRACT

We study the determinants of vertical integration in a new dataset of over 750,000 firms from 93 countries. Existing evidence suggests the presence of large cross-country differences in the organization of firms, which may be related to differences in financial development, contracting costs or regulation. We find cross-country correlations between vertical integration on the one hand and financial development, contracting costs, and entry barriers on the other that are consistent with these "priors". Nevertheless, we also show that these correlations are almost entirely driven by industrial composition; countries with more limited financial development, higher contracting costs or greater entry barriers are concentrated in industries with a high propensity for vertical integration. Once we control for differences in industrial composition, none of these factors are correlated with average vertical integration. However, we also find a relatively robust differential effect of financial development across industries; countries with less-developed financial markets are significantly more integrated in industries that are more human capital or technology intensive.

Daron Acemoglu Department of Economics MIT, E52-380B 50 Memorial Drive Cambridge, MA 02142-1347 and NBER daron@mit.edu

Simon Johnson Sloan School of Management MIT, E52 50Memorial Drive Cambridge, MA 02142-1347 and NBER sjohnson@mit.edu

Todd Mitton BYU todd.mitton@byu.edu

1 Introduction

Casual empiricism suggests the presence of significant differences in the organization of production across countries. For example, firms are often thought to be larger and more vertically integrated in less developed countries. Khanna and Palepu (1997, 2000) provide evidence consistent with this view and suggest that this is because market and contractual relationships are more costly in less-developed countries. Nevertheless, there has not been a systematic analysis of cross-country differences in vertical integration and their causes. Our primary aim in this paper is to make a first attempt at such a systematic analysis and to investigate the relationship between various institutional characteristics and vertical integration across countries.

Three well-established theories offer predictions on how differences in (specific) institutional characteristics of countries should affect vertical integration. First, according to the highly influential Transaction Cost Economics (TCE) theory pioneered by Williamson (1975, 1985), the internal organization of a firm is designed to improve incentives and limit agency costs. Vertical integration is perhaps the best known application of this theory. Vertical integration encourages specific investments and reduces holdup problems when markets are imperfect.¹ According to TCE, vertical integration should therefore be more prevalent when it is harder to write long-term contracts between upstream and downstream firms.

A second body of work emphasizes the importance of contracts and other relationships between firms and financial intermediaries. In this view, credit market imperfections affect the organization of the firm. Monitoring and contract enforcement are costly, so entrepreneurs need collateral in order to obtain financing (Banerjee and Newman, 1993, Legros and Newman, 1996), and they may need to rely on bank financing (Diamond and Rajan, 2005, Diamond, 2004). When credit markets have greater imperfections and when a lack of financial development limits the pool of potential entrepreneurs, there should be less entry and, most likely, larger firms in a country (Rajan and Zingales, 1998, Kumar, Rajan and Zingales, 1999). Larger firms are more likely to produce some of their own inputs or market some of their own outputs, so the financial view suggests that better financial institutions and credit markets may be associated with less vertical integration.

Third, recent work by Djankov, La Porta, Lopez-de-Silanes, and Shleifer (2002) and the

¹See the surveys by Holmstrom and Tirole (1989), Joskow (2005), and Whinston (2001). See also the related but different approach to vertical integration in the property rights theory developed by Grossman and Hart (1986) and Hart and Moore (1990), which points out both the benefits and the costs of vertical integration on incentives to undertake relationship-specific investments. Other important theoretical contributions in the area of vertical integration include Klein, Crawford and Alchian (1978), Bolton and Whinston (1992), Aghion and Tirole (1997), Baker, Gibbons, and Murphy (2002), and Legros and Newman (2003).

World Bank (2005) stresses the importance of regulatory barriers to entry. Ease of entry is also found to be related to firm size across countries (Klapper, Laeven, and Rajan, 2004). If there is less entry, presumably this makes vertically integrated firms more likely. Consequently, the regulation view suggests a relationship between entry regulations and vertical integration.

To investigate the cross-country determinants of vertical integration and the role of specific institutional features emphasized by these three theories, we use a new dataset of over 750,000 firms from 93 countries. Our methodology follows the finance literature in taking the United States as a benchmark (Rajan and Zingales, 1998), and we combine our firm-level data with the U.S. input-output tables (which is assumed to accurately describe the technological possibilities in other parts of the world). While there are some limitations to our data, it nonetheless provides a new opportunity to understand how the organization of production differs across countries. We have three sets of results.

First, we find cross-country correlations between measures of specific institutions emphasized by the theories and vertical integration. In particular, there is more vertical integration in countries with greater contracting costs between firms, as measured by indices of contract enforcement costs. Vertical integration is also higher in countries with less credit or greater credit market imperfections, as indicated by a lower level of credit market development (although this result is stronger in firm-level data than in country-level data). Finally, vertical integration is also higher in countries with greater barriers to entry, as measured by indices of the regulation of entry.

Second, however, we find that these cross-country differences in vertical integration are almost entirely accounted for by differences in industrial composition across countries. Once we control for differences in industrial composition, contracting costs, credit market development, and entry regulation have little explanatory power for differences in vertical integration. Thus, it is *not* the case that countries with greater contracting costs, credit market imperfections, or entry regulations tend to be more vertically integrated in a given sector. Rather, such countries tend to be concentrated in sectors that are naturally vertically integrated wherever they are in the world. We document that this lack of a relationship between these specific institutional features and vertical integration is highly robust.

To further explore this phenomenon, we create an index of the "vertical integration propensity" of each country based on its industrial composition and the natural tendency of each industry to vertically integrate (proxied by the vertical integration of that industry in the United States). Contracting costs, financial development, and entry regulation are all significantly correlated with a country's vertical integration propensity, and this difference in industrial composition explains the correlation between these measures and vertical integration in regressions that do not control for industrial composition. However, because countries with higher contracting costs, weaker financial development, and greater entry regulation are also typically poorer, it is again not possible to conclude that these specific institutional features are the cause of differences in vertical integration propensity. In fact, when we control for log GDP per capita, our specific measures of institutions lose significance; log GDP per capita is a more robust predictor of the vertical integration index and of a country's vertical integration propensity.²

Overall, we conclude from this set of results that measures of contracting costs, financial development and regulation have limited explanatory power for the level of vertical integration in a country. These results therefore shed some doubt on the importance of these specific institutional factors in accounting for cross-country patterns of vertical integration.

Nevertheless, our third set of results suggest that differences in financial development (and financial institutions) across countries have an important effect on vertical integration in the more human capital and technology-intensive sectors. We document this by looking at the effect of the interaction between financial development and industry characteristics (in particular, physical capital, human capital and technology intensity) on vertical integration. We find that even after controlling for industrial composition and for per capita GDP, greater credit market development is associated with less vertical integration in industries that are human capital and technology intensive. This suggests that a lack of financial development may be preventing efficient organization of production in relatively high-tech and high human capital industries, though we are not able to rule out other potential explanations for this differential effect. In any case, it has to be emphasized that even these interaction results cannot be interpreted as causal relationships, and it may be some other (omitted) characteristics that lead to the relationship between vertical integration and the interaction of industry characteristics and financial development.

Our paper relates to the existing literature in a number of ways. The comparative finance literature finds that countries with less financial development will tend not to develop in industries requiring greater external finance (Rajan and Zingales 1998), but has not investigated cross-country differences in vertical integration or in the internal organization of firms.³

 $^{^{2}}$ The effect of GDP per capita may capture the relationship between sectoral composition and the stage of development, or the effect of some "broader" institutional features related to the enforcement of property rights, state-society relations or political stability (e.g., Acemoglu, Johnson and Robinson, 2001, 2002).

³There is a large literature on vertical integration in specific industries in the United States, including Joskow's (1987) seminal paper on ownership arrangements in electricity generating plants, Stuckey's (1983) study of integration between aluminium refineries and bauxite mines, Monteverde and Teece's (1982) investigation of integration in the automobile industry, Masten's (1984) work on the aerospace industry, Ohanian's (1994) work on the pulp and paper industry, Klein's (1988) work on the Fisher Body and General Motors relationship, Baker and Hubbard's (2001, 2003) study of the trucking industry, Lerner and Merges' (1998) work on the biotech sector, and Chipty's (2001) paper on market foreclosure in the cable television industry. Woodruff's (2002) work on the Mexican footwear industry is the only paper we are aware of that provide a systematic study of vertical integration in a developing economy. Finally, Antràs (2003) studies the relationship between capital intensity and outsourcing using 23 U.S. industries.

Also related to our paper are cross-country comparative studies, including Bain (1966), Pryor (1972), Scherer (1973), Nugent and Nabli (1992), Kumar, Rajan and Zingales (1999), Desai, Gompers, and Lerner (2003), Fisman and Sarria-Allende (2004) and Klapper, Laeven, and Rajan (2003).⁴ These papers typically focus on concentration, firm size, and entry. Earlier papers use OECD data, while more recent papers use data from the Amadeus database for Western and Eastern Europe, or from the Worldscope database, which contains information only for relatively large publicly traded firms. Our dataset is, to the best of our knowledge, unique in allowing us to look at a relatively broad cross-section of countries and a large sample of firms, including both private and public companies and medium-size as well as large firms. In addition, none of these studies focuses on the internal organization of the firm or vertical integration.

The paper proceeds as follows. The next section describes the data used for the study. Section 3 reports our basic results on the relationship between vertical integration and specific measures of institutions, including results controlling for industrial composition and a series of robustness checks. Section 4 discusses how specific measures of institutions relate to the vertical integration propensity of countries. Section 5 reports interaction results. Section 6 concludes.

2 Data and Descriptive Statistics

Our firm-level data come from WorldBase. This database, compiled by Dun & Bradstreet for the primary purpose of providing business contacts, contains information on millions of public and private firms around the world. For each firm, WorldBase reports the 4-digit SIC code of the primary industry in which the firm operates, and the SIC codes of up to five secondary industries, listed in descending order of importance.⁵ WorldBase includes data for 213 different countries. We exclude 19 of these because they are not defined as countries in the World Bank on-line World Development Indicators database.⁶ In addition, because not all of the countries in WorldBase include reporting of secondary industries, our analysis is restricted to 93 countries for which this information is available.

⁴Another well-known approach, the market foreclosure theory, views vertical integration as a method of increasing monopoly power by downstream firms (e.g., Perry, 1978, Aghion and Bolton, 1987, Hart and Tirole, 1990, Ordover, Salop, and Saloner, 1990, and Chipty, 2001). We show that our results are robust to controlling for measures of antitrust regulations (as in Dutz and Hayri, 1999). However, because the available data on cross-country differences in antitrust regulation are more limited, we do not focus on antitrust issues in this paper.

⁵In the entire sample, approximately 64% of firms report one SIC code, 24% report two codes, 8% report three codes, 2% report four codes, 1% report five codes, and less than 1% report six codes. Note that we do not have the breakdown of sales by SIC for firms active in multiple industries.

⁶This excludes 15 non-independent territories, three independent countries below the World Bank size threshold, and one disputed territory. Taiwan is retained despite not being in the World Bank database.

Our sample consists of all firms in these countries in the September 2002 WorldBase file, with a maximum of 30,000 per country (a limit imposed due to cost constraints). For those countries with more than 30,000 firms, the 30,000 largest are selected, ranked by annual sales. We include firms from all industries, except those operating only in "wholesale trade" and "retail trade" (we explain this omission below). After these adjustments to the data, we have a base sample of 769,199 firms in 93 countries.

We use the benchmark input-output accounts published by the Bureau of Economic Analysis (BEA) to calculate the degree of vertical integration for each firm in our sample (see Lawson, 1997, for a discussion of the accounts). Our methodology follows the approach of Fan and Lang (2000).⁷ The input-output accounts report the dollar value of each input used to produce the output of 498 different industries in the U.S. economy. We use the 1992 input-output accounts because these are the most recently published at the 6-digit input-output (IO) code level. Input-output tables from the U.S. should be informative about input flows across industries, to the extent these are determined by technology. For example, in all countries, car makers need to obtain tires, steel, and plastic from plants specialized in the manufacture of those goods.⁸

We begin by matching the 4-digit SIC codes from each firm in our sample with the appropriate 6-digit IO code, using the BEA's concordance guide (see Lawson, 1997). Following Fan and Lang (2000), we exclude IO code 69.01 and 69.02 (wholesale and retail trade) from our analysis because the input-output classification system does not define these categories finely enough to allow meaningful vertical integration calculations – almost all 4-digit SIC codes between 5000 and 5999 map into just these two IO codes.

For every pair of industries, IO_i and IO_j , the input-output accounts allow us to calculate the dollar value of IO_i required to produce a dollar's worth of IO_j in the United States. This amount, which we call the vertical integration coefficient, VI_{ij} , represents the opportunity for vertical integration between IO_i and IO_j , i.e., when it is higher, there is more use of input *i* in the production of output *j*.

Using the full set of vertical integration coefficients (i.e., VI_{ij} for every IO_i and IO_j), we calculate a vertical integration index for each firm in our dataset. The index is denoted by v_{cif} for firm f in industry i in country c, and is defined as

$$v_{cif} = \frac{1}{|N_f|} \sum_{j \in N_f} V I_{ij},$$

 $^{^7\}mathrm{See}$ also Acemoglu, Aghion, Griffith and Zilibotti (2004) for an application of a similar methodology on UK data.

⁸The use of the same input-output table across countries is justified when all countries share the same technology frontier and when either all production functions are Leontief or there is factor price equalization. However, even when these stringent assumptions are not satisfied, we expect there to be a correlation in the input use patterns across countries.

where N_f is the set of industries in which firm f is active and $|N_f|$ denotes the number of these industries. In words, we first sum the VI_{ij} coefficients between the firm's primary industry and all industries in which the firm operates. This sum represents the dollar value of inputs from industries in which the firm operates that is required to produce one dollar's worth of the firm's primary output. We then create a similar index v_{cif} for secondary industries in which a firm operates. The vertical integration index is then the average of these sums for each firm, and as such represents the average opportunity for vertical integration in all lines of a business in which the firm is active.⁹

Across all 769,199 firms in our dataset, this index ranges from 0 (i.e., no vertical integration) to 53.5 (i.e., an average of 53.5 cents worth of the inputs required to produce one dollar's worth of output are produced by industries in which the firm operates).

For an example of how the vertical integration index is created, consider a Japanese auto maker in our data (primary code 59.0301) which also has two secondary sectors in the WorldBase data: automotive stampings (41.0201) and miscellaneous plastic products (32.0400). The VI_{ij} coefficients between these industries are as shown in the following table:

			Output (j)	
		Autos	Stampings	Plastics
	Autos	.0043	.0000	.0000
Input (i)	Stampings	.0780	.0017	.0000
	Plastics	<u>.0405</u>	<u>.0024</u>	<u>.0560</u>
	SUM	.1228	.0041	.0560

The table shows that, for example, the VI_{ij} coefficient for stampings to autos is 0.078, indicating that 7.8 cents worth of automotive stampings are required to produce a dollar's output of autos, and this automaker has the internal capability to produce those stampings. (Notice that industries have VI_{ij} coefficients with themselves; for example, miscellaneous plastic products are required to produce miscellaneous plastic products.) The bottom row shows the sum of the VI_{ij} for each industry, for example, 12.3 cents worth of the inputs required to make autos can be produced within this firm. The vertical integration index for this firm, v_{cif} , is then the average of the sums in the bottom row.¹⁰

We construct a country-level vertical integration index, denoted by v_c , by averaging all v_{cif} in the country.¹¹ In regressions using the country-level indices we weight the regression by the number of firms included in the average for each index (an approach analogous to performing firm-level regressions). In addition, we look directly at firm-level regressions. The first two

⁹We also conducted extensive robustness checks using only the primary (SIC) industry of each firm. The results are essentially unchanged.

¹⁰The index could also be constructed putting greater weight on the more important industries. While it seems natural to emphasize the primary industries in the index, WorldBase does not report sales breakdowns by industry, so the weightings would be somewhat arbitrary. We have constructed the index using different weighting schemes and find little difference in the results.

¹¹We use the simple average. Weighted averages (based on sales or employees) produce similar results.

rows of Panel A of Table 1 report descriptive statistics for the vertical integration index at the firm and country level.

Panel A of Table 1 also provides descriptive statistics for the other country-level measures we use as independent variables. Row 4 is the log of GDP per capita in 2000. Our GDP estimates are PPP adjusted and are taken from the World Factbook.¹² Row 5 reports log population, taken from World Bank data for the year 2000.

Rows 6 and 7 of Table 1 report our two primary measures of contracting institutions. Row 6 reports the cost of enforcing a commercial contract, i.e., between two firms, from the World Bank (2004) Doing Business dataset. This is the cost of enforcing a debt contract worth 50% of Gross National Income (GNI) per capita. The cost is measured as a percent of the amount of the debt contract. Row 7 reports the number of procedures needed to collect the same contract, again from World Bank (2004). This variable is emphasized in the underlying academic study, Djankov et al (2003).

In Rows 8 and 9 we report our two primary measures of credit market development. Our first credit market measure, in Row 8, directly captures the availability of external finance. This is the value of domestic credit provided to the private sector (as a percent of GDP), taken from World Bank data for the year 2000. This measure has been used frequently in other work (see, e.g., Rajan and Zingales, 1998). Row 9 reports our second measure, the cost of creating collateral, as a percent of GDP per capita. Property has to be registered before it can be used as collateral, and the analysis of the World Bank (2005) suggests this is an important component of the costs of borrowing from the banking system.

Rows 10 and 11 report our two primary measures of entry regulation. Row 10 presents entry costs by new firms in each country (as a percent of GDP per capita). These data are available for 61 countries in our sample. They are obtained from the World Bank (2004) and are constructed using the methodology in Djankov et al (2002). In Row 11 we report an alternative measure of entry barriers – the number of procedures needed for entry.

In robustness checks, we use other measures of contracting costs, credit market development, and entry barriers, with very similar results (see Appendix Table A3). Appendix Table A1 reports correlation coefficients of the country-level variables. The other rows of Panel A in Table 1 report summary statistics on the vertical integration propensity by country (Row 3, discussed in Section 4), and the number of employees per firm (Row 12). Panel B reports characteristics of relevant industries from U.S. data (discussed in Section 5).

¹²On the web at: http://www.cia.gov/cia/publications/factbook/. This covers a larger sample than the World Bank GDP estimates, and the two estimates are very similar for the countries for which they overlap.

3 Determinants of Vertical Integration

3.1 Country-level Results

According to theories emphasizing the role of contracting institutions in the internal organization of the firm, such as Williamson (1975, 1985), we should see a negative correlation between vertical integration and the quality of contracting institutions. Theories built on credit market constraints would suggest a negative association between vertical integration and credit market development. Finally, models of entry posit a relationship between vertical integration and entry barriers. Consequently, we would expect these variables to be correlated with cross-country differences in vertical integration. We investigate this question in Tables 2 and 3. Table 2 uses aggregate data, while Table 3 uses firm-level data.

Table 2 shows the relationship between aggregate indices of vertical integration and our specific measures of institutions. The following simple model is estimated by OLS:

$$v_c = \mathbf{x}_c' \boldsymbol{\beta} + \varepsilon_c, \tag{1}$$

where v_c is our index of aggregate vertical integration for country c calculated as described in the previous section, \mathbf{x}_c is a vector of country-level variables including the specific measures of institutions, and ε_c is an error term capturing all omitted factors. We do not interpret equation (1) as capturing a causal relationship, but as a convenient way of describing the correlation between specific measures of institutions and vertical integration around the world.

Columns 1 through 3 of Table 2 present results for our measures of contracting costs. Column 1 uses the cost of enforcing a contract. This variable is positive, with a coefficient of 1.72 and a standard error of 0.53. When contracting costs are higher, there is more vertical integration, as predicted by TCE. This relationship is illustrated in Figure 1, Panel A, which shows graphically the positive relationship between contracting costs and vertical integration. In Column 2 of Table 2 we add log population to control for country size, and the coefficient on contract enforcement cost decreases in magnitude, but remains significant. Column 3 uses the number of procedures needed to enforce a contract. The results are similar to those in Column 1 – the coefficient on contract enforcement procedures is positive and significant, as would be predicted by TCE. A one standard deviation reduction in the cost of enforcing contracts is associated with between 1/4 and 1/2 a standard deviation fall in the country-level vertical integration propensity, which is a large effect.

Columns 4 through 6 present results for our measures of credit market development. Columns 4 and 5 show that the coefficient on credit market development is negative, which suggests that stronger financial development is associated with less vertical integration. Nevertheless, the coefficient is not significant. This relationship is illustrated in Figure 1, Panel B, which shows a weak negative relationship between credit market development and vertical integration. Column 6 of Table 2 shows that the coefficient on cost to create collateral is neither significant nor of the expected sign. In the country-level regressions, there appears to be little evidence that credit market development is related to vertical integration.

Columns 7 through 9 present results for the two measures of entry regulation. Column 7 uses the cost of entry. This variable has a positive effect, with a coefficient of 1.28 and a standard error of 0.30. This positive relationship is consistent with priors; higher entry costs are associated with greater vertical integration, and the effect is large; a one standard deviation fall in entry costs is associated with about a 1/2 standard deviation decline in country-level vertical integration. Panel C of Figure 1 shows this significant positive relationship graphically. Column 8 show that this relationship holds when we control for log population, and Column 9 shows that the relationship holds with our second measure of entry regulation, the number of procedures required for entry.

3.2 Firm-level Results

In Table 3 we repeat the regressions of Table 2 using firm-level data.

A potential concern with the results in this paper is sample selection. Our dataset contains different numbers of firms from different countries, and this variation in the selection of samples of firms could be a source of variation in vertical integration. The main source of the problem would be potential correlation between vertical integration and firm size (combined with differential selection on firm size across countries). For example, it could be that relatively larger companies are more vertically integrated and from countries with weaker institutional environments we only observe relatively larger companies. We can partially deal with this sample selection problem by estimating our main equation at the firm level, and controlling for firm size. In other words, the estimating equation now becomes

$$v_{cf} = \mathbf{x}_c' \boldsymbol{\beta} + \mathbf{z}_f' \boldsymbol{\phi} + \varepsilon_{cf}, \qquad (2)$$

where v_{cf} is vertical integration in firm f in country c, \mathbf{x}_c is the set of country-level covariates as before, and \mathbf{z}_f is a set of firm-level covariates. Because the variables of interest, our specific measures of institutions, vary only at the country level, whenever we report regressions of this sort, the standard errors are corrected for clustering at the country level.

Table 3 reports the results from the estimation of (2) (with the log of the number of employees included as a measure of firm size).¹³ The general pattern is the same as in the country-level regressions in Table 2. On the whole, the results are somewhat stronger and more precise.

¹³We also experimented with regressions controlling for second, third and fourth order polynomials in firm size, and found very similar results (details available upon request).

Columns 1 through 3 again show that our measures of contracting costs are positively correlated with vertical integration, as would be predicted by TCE. The magnitude and significance of the coefficients is greater that the comparable estimates in Table 2 in all three columns. However, the standard deviation of the firm-level vertical integration index is much larger than the standard deviation of the country-level vertical integration index, so the implied effect of contracting costs on vertical integration is smaller (about 1/4 of the size in Table 2).

Columns 4 through 6 now report a statistically significant correlation between credit market development and vertical integration. In particular, the coefficient on domestic credit to the private sector is now -0.34 with a standard error of 0.15. As would be predicted, greater financial development is associated with less vertical integration. The measure of the cost to create collateral remains insignificant at the firm level.¹⁴

Columns 7 through 9 again show that our measures of entry regulation are positively correlated with vertical integration. With these measures, the firm-level results are somewhat smaller than the corresponding country-level results.

Overall, the country-level and firm-level regressions show that the measures of contracting institutions, credit market development, and regulation policy all appear to be correlated with the level of vertical integration in a country.

3.3 Industrial Composition

The results in Tables 2 and 3 documented the cross-country correlation between specific measures of institutions and the aggregate level of vertical integration. However, a missing element in our analysis thus far has been the lack of a control for differences in industrial composition across countries. It could be, for example, that countries with weaker institutional environments have economic activity concentrated in sectors that naturally have greater vertical integration. The simplest strategy to investigate whether industrial composition is an important concern is to include a full set of industry dummies in the firm-level regressions.

Consequently, the estimating equation becomes:

$$v_{cif} = \mathbf{x}_c' \boldsymbol{\beta} + \mathbf{z}_f' \boldsymbol{\phi} + \delta_i + \varepsilon_{cif}, \tag{3}$$

where v_{cif} is vertical integration of firm f in industry i of country c, \mathbf{x}_c and \mathbf{z}_f are countrylevel and firm-level covariates as before, and most importantly, the δ_i 's are a full set of industry dummies. The presence of the dummies enables the model to capture cross-industry differences in the technological or other determinants of vertical integration. The industry

 $^{^{14}}$ We also performed additional tests with accounting standards as the explanatory variable. While we find that better accounting standards are also associated with less vertical integration, the relationship is somewhat weaker statistically (a coefficient of -0.99 with a standard error of 0.55). This result may be attributed to the fact that the accounting standards variable is only available for 30 of our sample countries.

dummies are defined at the two-digit IO level, which results in a set of 76 dummy variables.¹⁵ As with the estimates of equation (2), we cluster the standard errors at the country level to take account of the fact that the key explanatory variables do not vary by firm (or industry).

The inclusion of a full set of industry dummies implies that in equation (3), all crosscountry comparisons are relative to the "mean propensity to integrate" in a particular industry. In other words, this regression looks at, for example, whether firms in a country with worse contracting institutions are more of vertically integrated relative to firms in a country with better contracting institutions in the *same* industry.

Table 4 reports the results from the estimation of equation (3) with the full set of industry dummies included. The striking result is that there is no longer a significant relationship between any of the measures of specific institutions and vertical integration; the exception is the cost of creating collateral, but this has the "wrong" sign. Evidently, the correlation between specific institutional factors and vertical integration depicted in Tables 2 and 3 was primarily due to differences in the industrial composition of production across countries. As a result, there is no evidence that, within a given industry, vertical integration is more prevalent in countries with greater contracting costs, weaker credit market development, or greater entry regulation.

More specifically, as we document in greater detail in the next section, countries with worse contracting institutions, more limited financial development and higher entry barriers are more concentrated in industries that have typically higher vertical integration, such as mining (ferrous and nonferrous), petroleum and gas, leather, fabrics, chemicals, apparel, and electronic components.

The lack of a correlation between our specific measures of institutions and vertical integration can be interpreted in different ways. One possibility is that our measures of specific institutions do not adequately capture cross-country differences in these factors. Naturally, the various proxies for contracting institutions, financial development, and regulation policy are imperfect and potentially measured with error. Nevertheless, in addition to the results of Tables 2 and 3, previous work shows that these indices do have significant information content, and are correlated with economic outcomes (see, e.g., Djankov et al, 2002, 2003). Thus the lack of correlation between these measures and vertical integration is unlikely to be driven entirely by measurement error.

A second interpretation is that, even if these institutional factors do not affect the degree to which a given firm chooses to vertically integrate, they have an impact on economic outcomes across countries by influencing industrial composition. We investigate this possibility in the next section.

¹⁵We use the primary industry of each firm, i.e., the IO code matched to the SIC code that comes first in WorldBase.

The final possibility is simply that these specific institutions have no impact on average vertical integration across countries. Such an interpretation would be a challenge to the theories discussed in the introduction, which (implicitly or explicitly) suggest that differences in contracting costs, credit markets, or regulation policy should have a major effect on cross-country patterns of vertical integration. We will see in Section 5 that this interpretation needs to be qualified; one of these factors, financial development, has a significant effect on vertical integration in the human capital and technology-intensive industries.

3.4 Robustness Checks

Before investigating the relationship between specific institutions and vertical integration further, we present a series of robustness checks of our results to this point. We verify that without controls for industry, the correlation between specific institutions and vertical integration is robust. In addition we show that the lack of a correlation between these variables after controlling for industry is also robust with two minor exceptions.

In Table 5 we present a series of three robustness checks. In this table we alternate columns of firm-level results without industry dummies, as in equation (2), with columns of firm-level results with industry dummies, as in equation (3). We present each robustness check only for our first proxy for each of the three types of institutions.

Panel A of Table 5 reports results for manufacturing industries only. We define "manufacturing" industries according to the BEA's classification, which means we exclude basic industries (such as agriculture and mining), and service industries (such as lodging, repair services, legal services, and health services), as well as transportation, communications, utilities, and finance-related industries. Columns 1 and 2 show that contracting costs remain significant in this sample, but lose significance once industry dummies are included. Columns 3 and 4 show that greater credit market development is associated with less vertical integration in this sample, and that the significance disappears when industry dummies are included. Columns 5 and 6 show that entry regulation remains significant in this sample, but once again not when industry dummies are included.

Panel B of Table 5 presents results excluding the most and least vertically integrated industries in the sample. The purpose of this robustness check is to assess if results are driven by a small number of industries that technologically have a high or low level of vertical integration. We rank the vertical integration of industries by estimating vertical integration coefficients (dummies) for each industry in a firm-level regression of vertical integration on industry dummies using only U.S. data. We use U.S. data since vertical integration patterns in the United States are likely to be most informative about the tendency of industries to vertically integrate their activities in an environment with relatively developed financial markets and contracting institutions (and relatively free entry). These estimated dummy coefficients are reported in Appendix Table A2 and the methodology is discussed further in the next section. The results in Panel B of Table 5 are similar to those in Panel A, and show that our findings are robust to this change. The exception is that the coefficient on domestic credit to the private sector is now slightly below standard levels of significance without industry dummies (Column 3).

Panel C of Table 5 presents results that include only industries that are present in 90% of the countries in our sample. The purpose here is to assess if our results are driven by industries that only appear in a small subset of countries in the sample. Panel C shows that this is not the case. The results are similar to those presented previously – the institutional measures have strong explanatory power when industry dummies are not included, but very little explanatory power when industry dummies are included. As in Panel B, this is only weakly true for the financial development measure, which is now not significant in Column 3.

In Table 6 we further check the robustness of our results by repeating the test using an alternative database. For this robustness check, we use the Worldscope database, which has been used extensively in the previous literature. It should be noted, however, that for the purposes of this paper, Worldscope is not as well suited to the investigation here as our primary data source, WorldBase. From WorldBase we get over 50 times the number of observations as in Worldscope, with data from roughly twice as many countries. In addition, WorldBase includes privately held and medium-sized firms, whereas Worldscope only includes large, publicly traded firms.

Panel A of Table 6 reports results of regressions similar to those reported previously, but with the methodology described earlier now repeated on the Worldscope database. Panel A of Table 6 shows that the results are similar using the Worldscope database. Column 1 shows that the coefficient on contracting costs is 1.82, very close to the 1.86 obtained in our baseline results. The standard error is higher than in the baseline results, perhaps because the number of countries included in the regression is fewer, so that the coefficient is slightly below standard levels of significance. Column 2 shows that, as in our baseline results, contracting costs have little explanatory power when industry dummies are included.

Column 3 of Panel A shows that the coefficient on credit market development is -1.27, with a standard error of 0.35, which is a considerably stronger result than in our baseline results. Furthermore, in contrast to our baseline results, Column 4 shows that although the coefficient on financial development is weakened when industry dummies are included, the coefficient remains significant and is one of the two exceptions to the general pattern of the significant effects disappearing once industry dummies are included.

Column 5 of Panel A shows that the coefficient on the entry cost is 2.16, with a standard error of 0.94, which is somewhat stronger than our baseline results. Again, Column 6 shows

the significance of the coefficient disappears when industry dummies are included.

Another concern with the results is that we have not so far incorporated information on business groups, which are important particularly in a number of Asian countries (see, for example, Khanna and Palepu, 2000). Panel B of Table 6 investigates this issue. We adjust the Worldscope data for group affiliations for Asian countries using data from Claessens et al (2000).¹⁶ In this adjustment, we treat all firms belonging to the same business group as a single entity, and aggregate their firm-level data accordingly.

Panel B reports the group-adjusted results. In general, the results are similar to the unadjusted results shown in Panel A. The coefficients on the institutional measures are greater than the corresponding measures in Panel A when industry dummies are not included. With industry dummies included, the coefficient on domestic credit to the private sector remains negative and marginally significant, though with a substantially smaller magnitude than in Panel A.

Additional robustness checks in Appendix Table A3 show that alternative measures of specific institutions are also correlated with vertical integration. The results are robust to measuring contracting costs as the time required to enforce a contract or the procedural complexity of contract enforcement (as in Djankov et al, 2003), to measuring credit market development as the disclosure index (from World Bank, 2005) and the interest rate spread between lending and borrowing rates (from World Bank data from 2000), and to measuring entry costs as the time required for entry (as in Djankov, et al, 2002). In addition, Table A3 shows that stronger antitrust regulation is associated with less vertical integration (with antitrust measured as in Dutz and Hayri, 1999). In all cases, there is a correlation between vertical integration in these specific institutions, and this correlation disappears when we include industry dummies (i.e., when we control for industrial composition).

4 Vertical Integration Propensity

We now investigate the reason why the significant correlation between vertical integration and our measures of specific institutions disappears when industry dummies are included in the regressions. For each country, we calculate the propensity to vertically integrate according to industrial composition:

$$\hat{V}_c = \sum_i \hat{\delta}_i \frac{S_{ci}}{S_c},$$

where $\hat{\delta}_i$'s are the estimates of the industry dummies (reported in Table A2) from a firm-level regression of vertical integration on industry dummies using U.S. data, S_{ci} is total sales in

¹⁶We are unable to do any such adjustments to our primary dataset because our data do not include firm names.

industry *i* in country *c*, and S_c is total sales in country *c*. The dummies $\hat{\delta}_i$'s measure the average level of vertical integration in industry *i* in the U.S., so \hat{V}_c measures the average tendency for vertical integration in the country due to its industrial composition. In other words, \hat{V}_c measures the extent of vertical integration in a country if the country had the average level of vertical integration in the United States corresponding to each industry.¹⁷ Consequently, the source of variation in \hat{V}_c arises purely from the industrial composition of the country.

In Table 7, we report results from regressions similar to those in equation (1), but with the vertical integration propensity of each country, \hat{V}_c , as the dependent variable. The explanatory variables are the specific measures of institutions as used previously. The results in Panel A of Table 7 show a strong correlation between specific measures of institutions and vertical integration propensity. Vertical integration propensity is significantly higher in countries with high contracting costs (Columns 1 and 2), in countries with high costs to create collateral (Column 4) and in countries with high entry regulations (Columns 5 and 6). Stronger credit market development is also associated with less vertical integration propensity, though this effect is not statistically significant (Column 3). In Column 7, we include all measures simultaneously. In this regression, contract enforcement costs and procedures required for entry emerge as having the most significant correlations with vertical integration propensity.

Figure 2 illustrates graphically the relationships reported in the regressions of Table 7. Panel A shows a strong positive relationship between contracting costs and vertical integration propensity, Panel B shows a negative (but not significant) relationship between credit market development and vertical integration propensity, and Panel C shows a strong positive relationship between entry costs and vertical integration propensity.

The results in Panel A of Table 7 illustrate why the correlation between vertical integration and specific measures of institutions disappears when we control for industrial composition (industry dummies). Countries with weaker institutions, as measured by contracting costs, credit market development, and entry regulation, tend to be concentrated in industries with a high technological propensity for vertical integration. Consequently, when we do not control for industrial composition, we see sizable differences in vertical integration across countries, but when we take into account of these differences in industrial composition, the correlations disappear.

How should these results be interpreted? On the one hand, the relationship between the specific institutional features and vertical integration propensity might itself reflect omitted

¹⁷In alternative tests we have also calculated the industry dummy coefficients using data from all G7 nations and also using data from all 93 countries in our dataset. The results are very similar to our baseline results and are available upon request.

factors.¹⁸ On the other hand, it is plausible that this relationship could be related to the effect of these specific institutional characteristics on the industrial composition of a country. The evidence we present next favors the former interpretation.

Panel B of Table 7 repeats the regressions of Panel A, but also including log GDP per capita as an additional explanatory variable. GDP per capita is a potential control both for the effect of the stage of development on industrial composition and may also capture the effect of other (broader) institutional features that are omitted from the regression. The results in Panel B indicate that the explanatory power of per capita GDP subsumes the explanatory power of the specific measures of institutions. In Column 1, for example, per capita GDP has a coefficient of -0.69, with a standard error of 0.14, whereas the coefficient on contract enforcement costs is insignificant. The results are similar in other columns; contract enforcement procedures is the only measure that is significant, but it has the opposite of the sign it had without GDP (negative rather than positive).

Panel B shows that the most robust relationship is that richer countries are more concentrated in industries with lower vertical integration propensity. The strong negative relationship between log GDP per capita and vertical integration propensity is shown graphically in Panel D of Figure 2.¹⁹

One concern with Table 7 is that the vertical integration propensity measure, \hat{V}_c , constructed using industrial composition of countries based on our primary data source, which may not be as representative of the overall industrial composition as some other international datasets. To address this concern, we repeat the analysis of Table 7, calculating \hat{V}_c using industrial composition calculated from an alternative data source. Of the alternative data sources available, the United Nations Industrial Development Organization (UNIDO) database is probably best suited to the task. While the country coverage is not as extensive as WorldBase, it is available for a large number of countries and offers good coverage of industrial composition in those countries. UNIDO does not offer firm-level data, but its industry-level data is sufficient for calculating \hat{V}_c when combined with our estimates of vertical integration coefficients, $\hat{\delta}_i$'s, from our WorldBase dataset.

Accordingly, we calculate the vertical integration propensity, \hat{V}_c , as defined above, but with industrial composition of countries, S_{ci}/S_c , calculated from the UNIDO database. Table 8 reports results of regressions using this measure of \hat{V}_c . Panel A shows results without log

¹⁸Potentially important among these omitted factors are broad institutional characteristics, related to property rights enforcement, corruption, state-society relations and political stability, as well as any effect on the stage of economic development on industrial composition. Appendix Table A1 shows that the correlation of per capita GDP with the specific institutional measures is generally high, suggesting that such omitted effects could be important.

¹⁹These results are not inconsistent with a view in which these specific measures of institutions indirectly affect vertical integration propensity through their effect on per capita GDP. Nevertheless, since many other factors affect GDP per capita, we do not find this view compelling.

GDP per capita. The results are generally similar to those in Panel A of Table 7. All our measures of specific institutions have coefficients of the same sign as in Table 7. All but two of the coefficients that were significant in Table 7 are significant in Table 8, a difference that may be attributable to the smaller number of countries included in the regressions in Table 8. Panel B of Table 8 shows that, again, per capita GDP dominates the specific measures of institutions in explaining vertical integration propensity. Though not always significant in this sample of countries, per capita GDP shows a strong negative correlation with a country's vertical integration propensity.

We further investigate the relationship between log GDP per capita and vertical integration in Table 9. Because, as Tables 7 and 8 show, per capita GDP has a strong correlation with vertical integration propensity, we would expect that per capita GDP would also be correlated with the vertical integration index. In Panel A of Table 9 we estimate firm-level regressions without industry dummies as in equation (2), and in Panel B we estimate regressions with industry dummies as in equation (3), and in each case per capita GDP is included as the main coefficient of interest. Panel A demonstrates a significant negative correlation between per capita GDP and vertical integration, which is consistent with the relationship between this variable and the vertical integration propensity (and is also consistent with "priors"). For example, in Column 1, the coefficient on per capita GDP is -0.48, with a standard error of 0.11, indicating that there is less vertical integration in richer countries. Columns 2 through 7 show that this relationship holds when we control for each of our measures of specific institutions. At the same time, the coefficients on the measures of specific institutions themselves are not significant. Panel A further demonstrates that the explanatory power of per capita income subsumes the explanatory power of specific institutional factors. Panel D of Figure 1 illustrates the relationship between per capita GDP and vertical integration graphically.

Despite the strong correlations in Panel A, Panel B of Table 9 shows that even this relationship does not hold when we control for industrial composition. In Panel B, the coefficient on log GDP per capita is now positive and generally not significant. Evidently, the entire correlation between GDP per capita and vertical integration is also due to industrial composition (or due to the relationship between GDP per capita and the vertical integration propensity, documented above).

Overall, we conclude that there is no evidence that any of the specific institutional features we have focused on or per capita income are systematically related to vertical integration once we take into account differences in industrial composition associated with the level of economic development.

5 Differential Effects Across Industries

The results in the previous sections may suggest that there are no robust regularities in cross-country vertical integration patterns once we control for industrial composition. In this section, we show that this is not entirely true by looking at the differential effects of these characteristics across industries.

The regression equations so far impose a "constant effect" of specific institutional characteristics on vertical integration. Another possibility is that these characteristics have differential effects across industries. For example, some industries may be systematically more vertically integrated in countries with weak institutions while other industries are more vertically integrated in countries with strong institutions. This might result, for instance, if both market transactions and contracting relationships within firms are more imperfect in poor countries, but also avoiding market imperfections are more important for some industries (leading to more vertical integration), whereas contracting problems make vertical integration more problematic for other industries.

5.1 Specifications with Interaction Effects

Motivated by these considerations, we estimate regressions of the following form

$$v_{cif} = \alpha y_c m_i + \beta x_c m_i + \mathbf{z}'_f \boldsymbol{\phi} + \delta_i + \eta_c + \varepsilon_{ci}.$$
(4)

where y_c represents (log) income per capita, x_c represents one of our measures of specific institutions, and m_i represents industry-level characteristics, such as capital intensity, human capital intensity, and technology intensity. The main effect for m_i is already taken out by the full set of industry dummies, the δ_i 's. The main coefficient of interest in this specification is β , and for this reason, we also include in this equation a full set of country dummies, η_c . The term $y_c m_i$ is included to investigate whether the interaction is between the specific institutional features and industry characteristics as opposed to some other factor related to income per capita (for example, a broader notion of institutional differences). As with all specifications that include interactions, all main effects are evaluated at their sample mean values. We also include firm-level characteristics, specifically a measure of firm size (as \mathbf{z}_f).

Estimates from equation (4) are reported in Table 10. Following the methodology in Rajan and Zingales (1998), all of the industry-level measures are based on U.S. data. In doing so we are assuming (analogous to assumptions made in Rajan and Zingales, 1998) that characteristics of industries in the U.S. economy are representative of (or at the very least correlated with) the characteristics of the same industries in other countries. Descriptive statistics for these measures are found in Panel B of Table 1.

We use three industry-level measures as interactions: physical capital intensity, human capital intensity, and the ratio of office and computing equipment to total equipment, which we refer to as "technology intensity" throughout. We take these measures from Autor, Katz and Krueger (1998), who calculate these using data from the National Income and Product Accounts (NIPA) for the year 1990. Physical capital intensity is defined as the log of capital investment to value added, human capital intensity is defined as the log of employees (full-time equivalent) to output, and as noted above, technology intensity is defined as the log of net capital stock in office, computing, and accounting machinery to total net capital equipment. We use the concordance that these authors developed to map the NIPA industries to our IO industries.

The interaction of log GDP per capita with the industry characteristic is not included in Panel A of Table 10, but is added in Panel B. Columns 1 through 3 present results with physical capital intensity as the industry characteristic. Panel A shows that the strongest effect is for credit market development interacted with capital intensity (Column 2). The coefficient of 0.42, with a standard error of 0.16, would suggest that firms in capital intensive industries are more likely to be vertically integrated in countries with greater financial development. However, in Panel B, including per capita GDP interacted with capital intensity eliminates the significance of credit market development interacted with capital intensity.

Columns 4 through 6 of Table 10 investigate the relationship between human capital intensity and vertical integration. In these columns, the strongest effect comes from the interaction between financial development and human capital or technology intensity. For example, in Column 5 of Panel A, the coefficient on the interaction term is -0.44, with a standard error of 0.18. This indicates that in countries with relatively strong financial development, vertical integration is less prevalent in human capital intensive industries. This effect is of a reasonable magnitude. For example, the coefficient estimate implies that in countries with the strongest financial development, a movement from the lowest human capital industry to the highest is associated with about a 1/4 standard deviation *fall* in the vertical integration index, whereas in countries with the weakest financial development, the same increase in human capital intensity is associated with a 1/5 standard deviation *rise* in the vertical integration index.

One concern is that this interaction effect, like the patterns in Table 7, may reflect an interaction between industry characteristics and GDP per capita. However, the estimates in Panel B indicate that the significant interaction between financial development and human capital intensity is robust to controlling for log GDP per capita. In fact, the coefficient hardly changes when we include the interaction between GDP per capita and industry characteristics.

Columns 7 through 9 of Table 10 look at the relationship between technology intensity and vertical integration. As with human capital intensity, in Panel A, the strongest effect is again on credit market development interacted with technology intensity, which has a negative and significant coefficient. Panel B shows that this result is robust to including log GDP per capita interacted with technology intensity. The coefficient on entry cost interacted with technology intensity is also significant in Panel B, but not in Panel A.

Overall, the interaction results suggest that credit market development is an important determinant of vertical integration in certain industries. More credit market development appears to reduce vertical integration in technology and human capital intensive industries.

In Table 11 we perform additional tests to assess the robustness of the relationship between vertical integration and the interaction of credit market development with human capital intensity and technology intensity. In Columns 1 and 2 of Table 11 we control for financial development interacted with each industry's dependence on external finance. Rajan and Zingales (1998) show that financial development has a more pronounced effect on growth in industries that are more dependent on external finance, so part of our results might be capturing the interaction between financial dependence and financial development. To control for this, we follow Rajan and Zingales (1998) and calculate each industry's technological dependence on external finance based on U.S. data in the CRSP database. Our measure follows the exact definition in Rajan and Zingales (1998) and is computed from U.S. data from 1990-1999. Columns 1 and 2 show that financial development interacted with external dependence on finance has very little impact on vertical integration, whereas the interactions of financial development with human capital intensity (Column 1) and technology intensity (Column 2) retain significant explanatory power. In Columns 3 and 4, we repeat the tests using an alternative measure of financial dependence from Rajan and Zingales (1998), the industry's dependence on equity (calculated from the same CRSP data). Again, this does not affect the significance of our interactions with human capital intensity and technology intensity. The final columns of Table 11 repeat the robustness tests of Table 5. These columns show that our interaction results are robust to excluding the most and least vertically integrated industries (Columns 7 and 8), and to including only industries appearing in at least 90% of the sample countries (Columns 9 and 10). Finally, the interactions continue to have the right sign but are no longer statistically significant when we restrict the regression to manufacturing industries (Columns 5 and 6), which suggests that differences in vertical integration in some high human capital service industries (such as electric and gas utilities, repair services, and amusements) are important for the interaction results.

Overall, the results of Tables 10 and 11 illustrate an interesting pattern relating financial development to vertical integration. They suggest that, when financial development is limited, the greatest extent of vertical integration is to be found in industries with advanced technology and greater than average human capital requirements.

5.2 Interpretation

How do we interpret the interaction effects reported in Tables 10 and 11? At face value, they suggest that while financial development has little effect on average vertical integration, it reduces vertical integration in technology-intensive and human capital-intensive sectors. This also naturally implies that financial development must have some positive effect on vertical integration in less technology-intensive sectors.

To interpret this pattern, let us return to the theories related to financial development. The crux of these theories is that lack of external (bank and market) finance will prevent entry of new firms and productive investment by existing firms. While we may expect that this will lead to larger and thus more vertically integrated firms, in fact the opposite might also be the case. For example, it may be efficient (either technologically or because of contractual reasons) for downstream firms to integrate with their upstream suppliers, but such a relationship may not emerge if downstream firms are credit constrained. On the other hand, if upstream firms are credit constrained and cannot undertake the necessary investments, vertical integration may occur even when it is not efficient.

In this light, the patterns we document are consistent with a configuration whereby in countries with limited financial development, low-tech sectors are insufficiently integrated, while high-tech sectors are excessively integrated. To investigate this issue further, we looked for evidence that the productivity implications of vertical integration are different depending on industry characteristics. In particular, we estimated models with productivity (sales per employee) on the left hand side and the triple interaction of financial development, the firmlevel vertical integration index and human capital intensity and also human capital intensity squared (or technology intensity and technology intensity squared) on the right hand side.²⁰ If both high vertical integration in the high-tech sectors and low vertical integration in the low-tech sectors of financially less-developed countries is inefficient, we may expect a nonmonotonic effect of the triple interaction (so that vertical integration in financially lessdeveloped countries is associated with lower productivity specifically in the sector with the lowest or highest technology or human capital needs). This exercise did not show a nonmonotonic pattern (results available upon request). This may reflect the fact that some other mechanism is responsible for the differential effects of financial development, or it may result from the crudeness of the productivity measures in the WorldBase dataset. Further investigation of the nature and cause of these differential effects is an area for future research.

 $^{^{20}}$ And naturally we also included all the second level interactions, so that the triple interaction does not capture their omitted effects.

6 Conclusion

In this paper, we studied the cross-country determinants of vertical integration in a new dataset of over 750,000 firms from 93 countries. Our focus was on the effect of specific institutional features on the vertical integration decisions of firms. This focus was motivated by both empirics and theory. Casual empiricism and existing work suggest that there are large differences in the organization of production and firms across countries and this may be related to contracting problems. Relatedly, a body of influential theories suggest that contracting costs (contracting institutions), credit market development and regulation should be important determinants of vertical integration.

Our empirical results do not confirm the main predictions of these theories. Although there is a cross-country correlation between vertical integration on the one hand and contracting costs, financial development, and entry barriers on the other, we show that this is entirely driven by industrial composition. In particular, countries with higher contracting costs or more limited financial development are concentrated in industries with a high propensity for vertical integration. Once we control for differences in industrial composition, none of these factors seem to affect vertical integration.

Nevertheless, our results also point to a significant relationship between financial development and vertical integration. We find a relatively robust differential effect of financial development across industries: countries with less-developed financial markets are significantly more integrated in industries that are more human capital or technology intensive.

We view our paper as a first step in understanding the cross-country patterns of organization of firms. Despite the importance of the organization of production for productivity and the existence of various influential theories, we know very little about these patterns. The dataset and the approach in this paper can be useful in investigating other dimensions of differences in the organization of firms across countries.

References

- Acemoglu, Daron, and Simon Johnson, 2003, Unbundling Institutions, NBER Working Paper #9934, Journal of Political Economy, forthcoming.
- [2] Acemoglu, Daron, Simon Johnson, and James Robinson, 2001, The Colonial Origins of Comparative Development: An Empirical Investigation, *American Economic Review* 91, 1369-1401.
- [3] Acemoglu, Daron, Simon Johnson, and James Robinson, 2002, Reversal of Fortune: Geography and Development in the Making of the Modern World Income Distribution, *Quarterly Journal of Economics* 117, 1231-1294.
- [4] Acemoglu, Daron, Philippe Aghion, Rachel Griffith, and Fabrizio Zilibotti, 2004, Vertical Integration and Technology: Theory and Evidence, NBER Working Paper #10997.
- [5] Aghion, Philippe, and Patrick Bolton, 1987, Contracts as a Barrier to Entry, American Economic Review 77, 388-401.
- [6] Aghion, Philippe, and Jean Tirole, 1997, Formal and Real Authority in Organizations, Journal of Political Economy 105, 1-29.
- [7] Antràs, Pol, 2003, Firms, Contracts, and Trade Structure, Quarterly Journal of Economics 118, 1375-1418.
- [8] Autor, David, Lawrence Katz, and Alan Krueger, 1998, Computing Inequality: Have Computers Changed the Labor Market?, *Quarterly Journal of Economics* 113, 1169-1214.
- [9] Bain, Joe, 1966, International Differences in Industrial Structure, Yale University Press, New Haven.
- [10] Baker, George, Robert Gibbons, and Kevin J. Murphy, 2002, Relational Contracts and the Theory of the Firm, *Quarterly Journal of Economics*, 117, 39-83.
- Baker, George, and Thomas Hubbard, 2001, Empirical Strategies in Contract Economics: Information and the Boundary of the Firm, *American Economic Review* 91, 189-194.
- [12] Baker, George, and Thomas Hubbard, 2003, Make Versus Buy in Trucking: Asset Ownership, Job Design, and Information, *American Economic Review* 93, 551-572.
- [13] Banerjee, Abhijit, and Andrew Newman, 1993, Occupational Choice and the Process of Development, *Journal of Political Economy* 101, 274-298.

- [14] Blanchard, Olivier and Michael Kremer 1997, Disorganization, Quarterly Journal of Economics, 112(4), 1091-1126.
- [15] Bolton, Patrick, and Michael Whinston, 1993, Incomplete Contracts, Vertical Integration, and Supply Assurance, *Review of Economic Studies* 60, 121-148.
- [16] Chipty, Tasneem, 2001, Vertical Integration, Market Foreclosure, and Consumer Welfare in the Cable Television Industry, *American Economic Review* 91, 428-453.
- [17] Claessens, Stijn, Simeon Djankov, Joseph Fan, and Larry Lang, 2000, The Separation of Ownership and Control in East Asian Corporations, *Journal of Financial Economics* 58, 81-112.
- [18] Desai, Mihir, Paul Gompers and Josh Lerner, 2003, Institutions, Capital Constraints and Entrepreneurial Firm Dynamics: Evidence from Europe, NBER Working Paper #10165.
- [19] De Soto, Hernando, 1989, The Other Path: The Invisible Revolution in the Third World, Harper and Row, New York.
- [20] Diamond, Douglas, 2004, Committing to Commit: Short-term Debt When Enforcement is Costly, *Journal of Finance* 59, 1447-1479.
- [21] Diamond, Douglas, and Raghuram Rajan, 2005, Liquidity Shortages and Banking Crises, Journal of Finance 60, 615-647.
- [22] Djankov, Simeon, Rafael La Porta, Florencio López-de-Silanes, and Andrei Shleifer, 2002, The Regulation of Entry, *Quarterly Journal of Economics* 117, 1-37.
- [23] Djankov, Simeon, Rafael La Porta, Florencio López-de-Silanes, and Andrei Shleifer, 2003, Courts, Quarterly Journal of Economics 118, 453-517.
- [24] Dollar, David, and Aart Kraay, 2003, Institutions, Trade, and Growth, Journal of Monetary Economics 50, 133-162.
- [25] Dutz, Mark, and Aydin Hayri, 1999, Does More Intense Competition Lead to Higher Growth?, World Bank Policy Research Working Paper #2320.
- [26] Easterly, William, and Ross Levine, 2003, Tropics, Germs and Crops: How Endowments Influence Economic Development, *Journal of Monetary Economics* 40, 3-39.
- [27] Fan, Joseph, and Larry Lang, 2000, The Measurement of Relatedmess: An Application to Corporate Diversification, *Journal of Business* 73, 629-660.

- [28] Fisman, Raymond, and Virginia Sarria-Allende, 2004, Regulation of Entry and the Distortion of Industrial Organization, Unpublished Manuscript, Columbia University.
- [29] Grossman, Sanford, and Oliver Hart, 1986, The Costs and Benefits of Ownership: A Theory of Vertical and Lateral Integration, *Journal of Political Economy* 94, 691-719.
- [30] Hart, Oliver, and John Moore, 1990, Property Rights and the Nature of the Firm, Journal of Political Economy 98, 1119-1158.
- [31] Hart, Oliver, and Jean Tirole, 1990, Vertical Integration and Market Foreclosure, Brookings Papers on Economic Activity Special Issue, 205-276.
- [32] Holmstrom, Bengt, and Jean Tirole, 1989, The Theory of the Firm, in *The Handbook of Industrial Economics*, Vol. I, Schmalensee, Richard, and Robert Willig, eds., Elsevier.
- [33] Joskow, Paul, 1987, Contract Duration and Relationship Specific Investments: Empirical Evidence from Coal Markets, American Economic Review 77, 168-185.
- [34] Joskow, Paul, 2005, Vertical Integration, in Handbook of New Institutional Economics, Ménard, Claude, and Mary Shirley, eds., Springer.
- [35] Khanna, T., and K. G. Palepu, 1997, Why Focused Strategies May Be Wrong for Emerging Markets, *Harvard Business Review* 75:4, 41-51.
- [36] Khanna, Tarun, and Krishna Palepu, 2000, Is Group Affiliation Profitable in Emerging Markets? An Analysis of Diversified Indian Business Groups, *Journal of Finance* 55, 867-891.
- [37] Klapper, Leora, Luc Laeven and Raghuram Rajan, 2004, Business Environment and Firm Entry: Evidence from International Data; NBER Working Paper #10380.
- [38] Klein, Benjamin, 1988, Vertical Integration as Organizational Ownership: The Fisher Body–General Motors Relationship Revisited, Journal of Law, Economics, and Organization 4, 199–213.
- [39] Klein, Benjamin, Robert Crawford, and Armen Alchian, 1978, Vertical Integration, Appropriable Rents, and the Competitive Contracting Process, *Journal of Law and Economics* 21, 297-326.
- [40] Kumar, Krishna, Raghuram Rajan, and Luigi Zingales, 1999, What Determines Firm Size?, NBER Working Paper #7208.
- [41] La Porta, Rafael, Florencio López-de-Silanes, Andrei Shleifer, and Robert Vishny, 1998, Law and Finance, Journal of Political Economy 106, 1113-1155.

- [42] Lawson, Ann, 1997, Benchmark Input-Output Accounts for the U.S. Economy, 1992, Survey of Current Business 77, 36-82.
- [43] Legros, Patrick, and Andrew Newman, 1996, Wealth Effects, Distribution and the Theory of Organization, *Journal of Economic Theory* 70, 312-341.
- [44] Legros, Patrick, and Andrew Newman, 2003, Competing for Ownership, CEPR Working Paper #2573.
- [45] Lerner, Josh, Merges, Robert, 1998, The Control of Strategic Alliances: An Empirical Analysis of Biotechnology Collaborations, *Journal of Industrial Economics* 46, 125-156.
- [46] Masten, Scott, 1984, The Organization of Production: Evidence from the Aerospace Industry, *Journal of Law and Economics* 27, 403-417.
- [47] Monteverde, Kirk and David Teece, 1982, Supplier Switching Costs and Vertical Integration in the Automobile Industry, *Bell Journal of Economics* 13, 206-213.
- [48] Nugent, Jeffrey, and Mustapha Nabli, 1992, Development of Financial Markets and the Size Distribution of Manufacturing Establishments: International Comparisons World Development 20, 1489-1500.
- [49] Ohanian, Nancy, 1994, Vertical Integration in the U. S. Pulp and Paper Industry 1900-1940, Review of Economics and Statistics 76, 202-207.
- [50] Ordover, Janusz, Steven Salop, and Garth Saloner, 1990, Equilibrium Vertical Foreclosure, American Economic Review 80, 127-42.
- [51] Perry, Martin, 1978, Price Discrimination and Vertical Integration, Bell Journal of Economics 9, 209-17.
- [52] Pryor, Frederic, 1972, An International Comparison of Concentration Ratios, *The Review of Economics and Statistics* 54, 130-140.
- [53] Raddatz, Claudio, 2005, Liquidity Needs and Vulnerability to Financial Underdevelopment, *Journal of Financial Economics*, forthcoming.
- [54] Rajan, Raghuram, and Luigi Zingales, 1998, Financial Dependence and Growth, American Economic Review 88, 559-586.
- [55] Rodrik, Dani, Arvind Subramanian, and Francesco Trebbi, 2004, Institutions Rule: The Primacy of Institutions over Geography and Integration in Economic Development, *Jour*nal of Economic Growth 9, 131-165.

- [56] Scherer, Frederic, 1973, The Determinants of Industrial Plant Sizes in Six Nations, The Review of Economics and Statistics 55, 135-145.
- [57] Stuckey, John, 1983, Vertical Integration and Joint Ventures in the Aluminum Industry, Harvard University Press, Cambridge.
- [58] Whinston, Michael, 2001, Assessing the Property Rights and Transaction-Cost Theories of Firm Scope, American Economic Review Papers and Proceedings 91, 184-188.
- [59] Williamson, Oliver, 1975, Markets and Hierarchies: Analysis and Antitrust Implications, Free Press, New York.
- [60] Williamson, Oliver, 1985, *The Economic Institutions of Capitalism*, Free Press, New York.
- [61] Woodruff, Christopher, 2002, Non-contractible Investments and Vertical Integration in the Mexican Footwear Industry, International Journal of Industrial Organization 20, 1197-1224.
- [62] World Bank, 2004, World Development Report: Making Services Work for Poor People, The World Bank and Oxford University Press, Washington and New York.
- [63] World Bank, 2005, World Development Report: A Better Investment Climate for Everyone, The World Bank and Oxford University Press, Washington and New York.





Чſ

Ë

MY S

NLUSA

LUX

PAN

CHN

MAC

THA DEU GBR









- 5

9

8 9 log GDP per capita

 \sim

ဖ

έ

BRAR FANATT FANATT DNK

CRI PWANP

S

7

ADO

Panel C: Vertical integration and entry costs





2

1.5

R

ЩIJ



Panel D: Vertical integration propensity and per-capita GDP

7

ASU CY MBMU

	Mean	25th Pctile	Median	75th Pctile	St. Dev.	z
Panel A: Key variables						
(1) Vertical integration index, firm level	4.87	0.46	3.34	6.79	5.58	769,199
(2) Vertical integration index, country level	4.98	4.38	4.93	5.50	1.00	93
(3) Vertical integration propensity	-11.81	-12.33	-11.84	-11.30	0.96	93
(4) GDP per capita (log)	9.03	8.21	9.17	9.98	1.00	93
(5) Population (log)	14.95	12.39	15.52	16.96	2.75	93
(6) Contract enforcement cost (% of debt)	0.26	0.11	0.18	0.30	0.26	61
(7) Contract enforcement procedures (/10)	2.80	2.20	2.70	3.50	0.94	61
(8) Credit market development (% GDP)	0.54	0.29	0.54	0.87	0.43	73
(9) Credit collateral cost (% GDP per capita)	0.13	0.01	0.05	0.16	0.23	55
(10) Entry cost (% GDP per capita)	0.38	0.09	0.17	0.47	0.49	61
(11) Entry procedures (/10)	0.88	0.70	0.90	1.10	0.35	61
(12) Number of employees (log)	3.86	2.56	3.74	5.00	1.87	676,046
Panel B: U.S. industry characteristics						
(13) Capital intensity (log)	-0.72	-1.00	-0.86	-0.58	0.55	67
(14) Human capital intensity (log)	-4.80	-5.11	-4.71	-4.55	0.49	67
(15) Technology intensity (log)	-2.96	-3.64	-2.80	-2.09	1.00	67
The table presents descriptive statistics of data used in s	ubsequent tabl	les. Panel A	presents fin	rm-level and e	country-level	data used in
regression analysis. Vertical integration measures and nu	mber of emple	yees come fi	rom the Sep	ot. 2002 World	lBase databa	se. GDP per
capita comes from the 2000 World Factbook. Population	and credit ma	arket develop	ment come	from 2000 W	orld Bank d	ata. Contract
enforcement cost, contract enforcement procedures, cred	it collateral co	ost, entry cos	t, and entry	r procedures a	re based on	World Bank
(2004, 2005) and Djankov et al. (2002, 2003). Panel B	presents U.S.	industry char	acteristics 1	used as interac	cation terms	in Section 5.
Industry characteristics come from Autor, Katz, and Krueg	ger (1998).					

Table 1 Descriptive statistics

Dependent variableContract enforcement cost1.720.89Contract enforcement procedures(0.53)(0.36)Contract enforcement procedures0.241Credit market development(0.116)Credit collateral cost-0.08Entry costEntry costEntry procedures0.32Log population0.32Log population0.32	Jant	(5)	(9)	(2)	(8)	(6)
Contract enforcement cost1.720.89Contract enforcement procedures(0.53)(0.36)Credit market development0.241(0.116)Credit market development0.241(0.116)Credit collateral cost11Entry cost11Entry procedures0.32Log population0.32Log population0.32	aent variable L	s the vertica	ıl integrat	ion index		
(0.53)(0.36)Contract enforcement procedures0.241Credit market development0.116)Credit collateral cost-0.08Credit collateral cost-0.08Entry cost-1.00Entry cost-1.00Log population0.32Log population0.32(0.06)-1.00			I			
Contract enforcement procedures 0.241 (0.116) Credit market development 0.116) Credit collateral cost Entry cost Entry cost Entry procedures 0.32 Log population 0.32 (0.06)						
Credit market development(0.116)Credit market development-0.08Credit collateral cost(0.18)Entry costEntry costEntry procedures0.32Log population0.32(0.06)(0.06)	- <					
Credit market development -0.08 Credit collateral cost Entry cost Entry procedures 0.32 Log population 0.32 (0.06)	(0)					
Credit collateral cost(0.18)Entry costEntry proceduresEntry procedures0.32Log population(0.06)	-0.08	-0.11				
Credit collateral cost Entry cost Entry procedures Log population 0.32 (0.06)	(0.18)	(0.17)				
Entry cost Entry procedures Log population 0.32 (0.06)			-0.36			
Entry cost Entry procedures Log population 0.32 (0.06)			(0.98)			
Entry procedures Log population 0.32 (0.06)				1.28	0.83	
Entry procedures Log population 0.32 (0.06)				(0.30)	(0.24)	
Log population 0.32 (0.06)						0.89
Log population 0.32 (0.06)						(0.34)
(0.06)		0.21			0.33	
		(0.10)			(0.06)	
R-squared 0.15 0.50 0.08 0.00	0.00	0.22	0.00	0.13	0.51	0.17
Number of Observations 61 61 73	73	73	55	61	61	61

(--. lotio. ---Table 2 dit market devi inctitutio . -7 .4 Vertical ir 4 ب the vertical integration index for each country. i

Vertical integration and	contracting i	nstitutions,	Table 3 credit mar	ket develo	pment, and	entry regu	ılation (firn	n level)	
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)
			Dependen	t variable i	s the vertic	al integrat	tion index		
Contract enforcement cost	1.86 (0.44)	1.34	ı			I			
Contract enforcement procedures		(00.0)	0.28						
Credit market development			(01.0)	-0.34	-0.32				
Credit collateral cost				(01.0)	(01.0)	0.00			
Entry cost						(10.0)	1.42	1.14	
							(0.26)	(0.23)	
Entry procedures									0.78 (0.25)
Log population		0.20 (0.06)			0.22 (0.07)			0.22 (0.06)	
Log number of employees	0.26 (0.05)	(0.04)	0.27 (0.05)	0.27 (0.05)	0.19	0.27 (0.05)	0.26 (0.05)	0.19	0.25 (0.04)
R-squared	0.01	0.01	0.01	0.01	0.01	0.01	0.01	0.01	0.01
Number of Observations	661,747	661,747	661,747	648,157	648,157	661,038	661,747	661,747	661,747
The table presents coefficient estimate Robust standard errors, adjusted for cl	es from regre ustering with	sssions of a	a firm-leve es, are in pa	l vertical in arentheses.	ntegration i	ndex on sl	pecific mea	ısures of ir	stitutions.

	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)
			Dependen	it variable i	is the vertic	cal integrat	tion index		
Contract enforcement cost	-0.09 (0.45)	-0.14 (0.47)							
Contract enforcement procedures			0.02 (0.07)						
Credit market development			·	0.02 (0.13)	0.02 (0.13)				
Credit collateral cost				~		-0.011 (0.004)			
Entry cost						~	0.12	0.10	
							(0.25)	(0.24)	
Entry procedures									-0.04 (0.18)
Log population		0.02 (0.04)			0.01 (0.03)			0.01 (0.04)	
Log number of employees	0.05 (0.03)	0.04 (0.03)	0.05 (0.03)	0.04 (0.03)	0.04 (0.03)	0.04 (0.03)	0.05 (0.03)	0.04 (0.03)	0.05 (0.03)
R-squared	0.40	0.40	0.40	0.40	0.40	0.40	0.40	0.40	0.40
Number of Observations	661,747	661,747	661,747	648,157	648,157	661,038	661,747	661,747	661,747
The table presents coefficient estimate Robust standard errors adjusted for ch	es from regre	ssions of a	a firm-leve es are in n	l vertical i	ntegration Also estir	index on sj nated but n	pecific mea	asures of ir	stitutions.
dummies defined at the two-digit input	t-output leve	Ï							

÷	evel)
۔ د	tirm I
•	egulation (
	id entry re
	an
-	development,
	G
-	mark
	credit
•	institutions,
•	contracting
-	and
•	integration
-	a
·	1 C

Table 4

	(1)	(2)	(3)	(4)	(5)	(6)
	Depe	endent vari	able is the	vertical in	tegration in	ndex
Panel A: Manufacturing industries onl	y 115	0.00				
Contract enforcement cost	1.16	0.08				
Credit market development	(0.31)	(0.48)	0.50	0.11		
Credit market development			-0.50	(0.12)		
Entry cost			(0.10)	(0.12)	1.05	0.20
					(0.28)	(0.28)
Log population	0.16	0.01	0.21	0.02	0.18	0.01
	(0.05)	(0.04)	(0.05)	(0.04)	(0.05)	(0.04)
Log number of employees	0.26	0.17	0.29	0.18	0.26	0.17
	(0.04)	(0.03)	(0.04)	(0.03)	(0.04)	(0.03)
Industry dummies	No	Yes	No	Yes	No	Yes
R-squared	0.01	0.40	0.01	0.39	0.01	0.40
Number of Observations	253,614	253,614	239,093	239,093	253,614	253,614
Panel B: Most and least vertically integ	rated indu	stries excl	uded			
Contract enforcement cost	1.00	-0.09				
	(0.31)	(0.46)				
Credit market development			-0.20	-0.03		
			(0.16)	(0.15)	0.05	0.11
Entry cost					0.85	(0.25)
Log population	0.16	0.01	0.17	0.00	(0.23)	(0.23)
Log population	(0.10)	(0.01)	(0.05)	(0.00)	(0.05)	(0.00)
Log number of employees	(0.04) 0.17	(0.04)	0.17	(0.04)	0.17	(0.04)
Log number of employees	(0.03)	(0.03)	(0.04)	(0.03)	(0.03)	(0.03)
Industry dummies	No	Yes	No	Yes	No	Yes
P squared	0.01	0.32	0.01	0.31	0.01	0.32
Number of Observations	544 067	0.32 544 067	531 300	531 300	544 067	0.32 544 067
Panel C: Industries appearing in at leas	st 90% of a	countries o	only	551,500	511,007	511,007
Cost to enforce a contract	1.46	-0.14				
	(0.46)	(0.47)				
Domestic credit to the private sector			-0.35	0.04		
			(0.20)	(0.15)		
Cost of entry					1.23	0.10
					(0.31)	(0.25)
Log population	0.23	0.02	0.24	0.01	0.25	0.02
	(0.08)	(0.04)	(0.08)	(0.04)	(0.08)	(0.04)
Log number of employees	0.16	(0.04)	0.17	(0.03)	0.16	(0.04)
	(0.05)	(0.03)	(0.05)	(0.03)	(0.04)	(0.03)
Industry dummies	No	Yes	No	Yes	No	Yes
R-squared	0.01	0.39	0.01	0.38	0.01	0.39
Number of Observations	567,179	567,179	560,644	560,644	567,179	567,179

Table 5	
Robustness checks	(firm level)

The table presents coefficient estimates from regressions of a firm-level vertical integration index on specific measures of institutions. Robust standard errors, adjusted for clustering within countries, are in parentheses. In specified columns, industry dummies, defined at the two-digit input-output level, are also estimated but not reported. In Panel B, "most and least vertically industries" includes the 5% most and 5% least vertically integrated industries.

	(1)	(2)	(3)	(4)	(5)	(9)
	Depe	ndent vari	able is the	vertical in	tegration in	ndex
Panel A: Unadjusted data	4)	
Contract enforcement cost	1.82	0.65				
	(0.93)	(0.72)				
Credit market development			-1.27	-0.81		
			(0.35)	(0.24)		
Entry cost					2.16	1.07
Log population	0.05	-0.04	0.30	0.08	0.05	-0.04
•	(0.13)	(0.0)	(0.12)	(0.08)	(0.14)	(0.00)
Log number of employees	0.04	0.04	0.02	0.03	0.03	0.04
	(0.05)	(0.04)	(0.05)	(0.04)	(0.05)	(0.04)
Industry dummies	No	Yes	No	Yes	No	Yes
R-squared	0.00	0.28	0.01	0.28	0.00	0.28
Number of Observations	9,914	9,914	9,596	9,596	9,914	9,914
Panel B: Group-adjusted data						
Contract enforcement cost	2.73	-0.38				
	(1.25)	(0.95)				
Credit market development			-1.47	-0.40		
			(0.38)	(0.20)		
Entry cost					2.50	0.62
					(1.02)	(0.63)
Log population	0.04	0.05	0.32	0.10	0.06	0.04
	(0.15)	(0.07)	(0.12)	(0.08)	(0.16)	(0.07)
Log number of employees	0.04	0.03	0.03	0.03	0.04	0.03
	(0.05)	(0.03)	(0.05)	(0.04)	(0.05)	(0.03)
Industry dummies	No	Yes	No	Yes	No	Yes
R-squared	0.00	0.35	0.01	0.36	0.00	0.35
Number of Observations	8,836	8,836	8,545	8,545	8,836	8,836
The table presents coefficient estimates	from regr	essions of	a firm-lev	el vertical	integration	i index on
specific measures of institutions. Data c	come from	n the Worl	dscope dat	abase. Ro	bust stand	ard errors,
adjusted for clustering within countries,	are in pa	rentheses.	In specifi	ed column	s, industry	dummies,

Table 6 Rohustness checks with Worldscone data (firm level)

defined at the two-digit input-output level, are also estimated but not reported. In Panel B, "Group-adjusted" means that Asian firms belonging to the same business group are treated as a single entity.

	(1)	(2)	(3)	(4)	(2)	(9)	(2)
		Dependent	variable i	s vertical i	ntegration]	propensity	
Panel A: Per-capita GDP not included Contract enforcement cost	1.87 (0.73)						1.53 (0.68)
Contract enforcement procedures	~	0.28 (0.11)					-0.01
Credit market development			-0.29				0.09
Credit collateral cost			((110)	1.87			(0.21) 1.35 (0.84)
Entry cost				(10.0)	1.43 (0.36)		-0.21 -0.21
Entry procedures						1.11 (0.25)	(0.38) (0.38)
R-squared	0.20	0.11	0.04	0.06 22	0.17	0.30	0.42
Number of Observations	61	61	73	55	61	61	53
Panel B: Per-capita GDP included							
Log GDP per capita	-0.69 (0 14)	-0.92 (0.12)	-0.79	-0.69 (0.12)	-0.72	-0.59 (0 14)	-0.89 (0.16)
Contract enforcement cost	-0.02						0.23
Contract enforcement procedures		-0.29					-0.26 (0.12)
Credit market development			0.36				0.29
Credit collateral cost			((110)	0.05			(0.20) 1.14 (0.83)
Entry cost				(07.0)	-0.20		(0.0) -0.28
Entry procedures					(0.30)	0.32 (0.24)	(0.34) 0.32 (0.27)
R-squared	0.52	0.58	0.58	0.52	0.52	0.53	0.69
Number of Observations	61	61	73	55	61	61	53
The table presents coefficient estimates fintegration on specific measures of instancial technology weighted by the number of firms ner court	rom regre titutions.	ssions of a Robust st	n index m tandard er	easuring a rors are ir	country's p	ropensity f	or vertical ssions are
weighted by the manifest of thins per cour	. y						

Table 8	Robustness check of vertical integration propensity with UNIDO data (country level)
---------	---

	(1)	(2)	(3)	(4)	(5)	(9)	(7)
		Dependent	variable i	s vertical in	ntegration]	propensity	
Panel A: Per-capita GDP not included							
Contract enforcement cost	2.51 (0.81)						1.58 (1.01)
Contract enforcement procedures	~	0.28 (0.09)					-0.12
Credit market development			-0.31				0.17
Credit collateral cost			(01.0)	1.71			(0.94) 3.09
Entry cost				(17:1)	0.86		0.69)
Entry procedures						0.42 (0.24)	0.52 (0.46)
R-squared	0.20	0.34	0.09	0.12	0.09	0.10	0.52
Number of Observations	28	28	30	27	28	28	27
Panel B: Per-capita GDP included							
Log GDP per capita	-0.41	-0.33	-0.32	-0.49	-0.70	-0.48	-0.52
Contract enforcement cost	0.99 (0.83)	(01.0)	(77.0)			(61.0)	0.88 0.88 (1.28)
Contract enforcement procedures		0.13 (0.13)					-0.17
Credit market development			-0.07				0.39
Credit collateral cost				0.02			2.62
Entry cost				(71.1)	-1.01		(C1.1) 0.11 (0.70)
Entry procedures					(0.40)	0.04 (0.24)	(0.58) 0.33 (0.62)
R-squared	0.40	0.41	0.16	0.39	0.44	0.38	0.55
Number of Observations	28	28	30	27	28	28	27
The table presents coefficient estimates finite integration on specific measures of insti-	rom regre tutions.	essions of a The proper	n index me nsity for ve	easuring a ertical integ	country's p gration is (ropensity f calculated	or vertical using data
from the UNIDO database. Kobust stand	ard errors	s are in pare	entheses. I	cegressions	s are weigh	ited by the	number of

firms per country.

	(1)	(2)	(3)	(4)	(2)	(9)	(1)
		Depend	ent variable	is the vertic	al integratio	n index	
Panel A: No industry dummies							
Log GDP per capita	-0.48	-0.40	-0.51	-0.55	-0.52	-0.42	-0.41
	(0.11)	(0.13)	(0.15)	(0.12)	(0.14)	(0.15)	(0.15)
Contract enforcement cost		0.72					
Contract anticarcount and and and		(00.0)					
Contract enforcement procedures			-0.02 (0.13)				
Credit market development				0.07			
Credit collateral cost					-1.09		
Entry cost					(61.0)	0.45	
Entry procedures						(0+.0)	0.26
Log number of employees	0.25	0.25	0.25	0.24	0.24	0.25	(0.26) 0.25
	(0.04)	(0.04)	(0.04)	(0.04)	(0.04)	(0.04)	(0.04)
R-squared	0.01	0.01	0.01	0.01	0.01	0.01	0.01
Number of Observations	676,046	661,747	661,747	648,157	661,038	661,747	661,747
Panel B: Industry dummies included							
Log GDP per capita	0.08	0.14	0.22	0.14	0.06	0.19	0.14
	(0.08)	(0.09)	(0.12)	(0.09)	(0.08)	(0.0)	(0.09)
Contract enforcement cost		0.29 (0.41)					
Contract enforcement procedures			0.14				
Credit market development			(60.0)	-0.08			
				(0.16)	10.0		
Credit collateral cost					10.0- (00.0)		
Entry cost						0.54	
Entry procedures						(+7.0)	0.12
I og munher of emulovees	0.05	0.05	0.06	0.05	0.04	0.05	(0.23) 0.05
	(0.03)	(0.02)	(0.02)	(0.03)	(0.02)	(0.02)	(0.02)
R-squared	0.38	0.40	0.40	0.40	0.40	0.40	0.40
Number of Observations	676,046	661,747	661,747	648,157	661,038	661,747	661,747
The table presents coefficient estimates 1 institutions Robust standard errors adjuit	from regressi	ions of a fir tering within	m-level ver	tical integra	tion index c	in specific in estimated	in Danel B
insulutions. Rooust standard errors, auju	sted tot cius	nun guna	n countries,	are in parer	ILLICSCS. AIS	o esumateu	in ranei d,

Table 9 Vertical integration and GDP per capita (firm level)

but not reported, are a set of industry dummies defined at the two-digit input-output level.

	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)
Industry characteristic interacted:	Ca	pital intens	sity	Huma	n capital in	tensity	Tech	nology inte	insity
			Depender	nt variable	is the vertic	al integrati	ion index		
Panel A: Per-capita GDP × industry	cnaracterist	ic not incli	<i>uaea</i>						
Contract enforcement cost	-0.62			0.41			-0.15		
	(+0.0)			(nc.n)			(07.0)		
Credit market development		0.42 (0.16)			-0.44 (0.18)			-0.27 (0.12)	
Entry cost			-0.43		a.	0.22		e.	-0.12
I og numher of emulovises	0.04	0.04	(0.23) 0.04	0.04	0.04	(0.25) 0.04	0.04	0.04	(0.15) 0.04
	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)
R-squared	0.43	0.42	0.43	0.43	0.42	0.43	0.43	0.42	0.43
Number of Observations	557,374	543,558	557,374	557,374	543,558	557,374	557,374	543,558	557,374
Panel B: Per-capita GDP × industry	characterist	ic includea	1						
Log GDP per capita	0.25	0.11	0.27	-0.20	0.00	-0.23	-0.13	0.04	-0.13
	(0.08)	(0.10)	(0.10)	(0.12)	(0.13)	(0.14)	(0.07)	(0.07)	(0.07)
Contract enforcement cost	0.08 (0.38)			-0.13 (0.38)			-0.48 (0.25)		
Credit market development		0.35 (0.20)			-0.44 (0.22)			-0.30 (0.14)	
Entry cost		~	0.21		~	-0.29		~	-0.39
	000	100	(0.33)		200	(0.35)	000	50.0	(0.19)
Log number of employees	0.04 (0.02)								
R-squared Number of Observations	0.43 557.374	0.42 543,558	0.43 557.374	0.43 557.374	0.42 543.558	0.43 557.374	0.43 557.374	0.42 543.558	0.43 557.374
The table presents coefficient estimat	es from regi	ressions of	a firm-lev	el vertical	integration	index on s	specific me	asures of i	nstitutions
interacted with U.S. industry character	ristics. Indu	stry charac	cteristics ar	e taken froi	m Autor, K	atz, and K	rueger (199	98). Robus	t standard
errors, adjusted for clustering within co	ountry/indus	tries, are ir	n parenthes	es. Also es	timated but	not reporte	ed are a set	of industry	/ dummies
defined at the two-digit input-output le	vel, and a se	t of countr	y dummies						

	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)
							Most a	nd least	Industries	appearing
	Control fo	r external	Control f	or equity	Manufa	loturing	vertically	integrated	in at leas	t 90% of
	depen	dence	depen	dence	industri	ies only	industries	excluded	countri	es only
			Dep	vendent var	iable is the	vertical int	egration ind	lex		
Credit market development × human capital intensity	-0.44		-0.44		-0.21		-0.383		-0.45	
	(0.21)		(0.18)		(0.23)		(0.197)		(0.19)	
Credit market development × technology intensity		-0.302		-0.27		-0.05		-0.28		-0.30
		(0.153)		(0.12)		(0.11)		(0.13)		(0.13)
Credit market development × external dependence	0.00	-0.09								
	(0.12)	(сг.0)								
Credit market development × equity dependence			0.06	0.08						
			(0.05)	(0.05)						
Log number of employees	0.04	0.04	0.04	0.04	0.19	0.19	0.05	0.06	0.03	0.03
	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)
R-squared	0.42	0.42	0.42	0.42	0.40	0.40	0.33	0.33	0.41	0.41
Number of Observations	543,202	543,202	543,202	543,202	239,093	239,093	432,681	432,681	464,531	464,531
The table presents coefficient estimates from regres	sions of a f	irm-level v	ertical inte	gration inc	ex on crea	dit market	developme	nt interacte	d with U.S	industry.
characteristics. Human capital intensity and technolog	y intensity co	me from A	utor, Katz,	and Kruege	r (1998). 1	External dej	pendence a	nd equity de	spendence &	tre defined
as in Rajan and Zingales (1998). "Most and least ve	rtically indus	stries" inclu	des the 5%	most and	5% least v	rertically in	tegrated in	dustries. R	obust stand	ard errors,
adjusted for clustering within country/industries, are in	I parentheses.	Also estin	nated but no	ot reported	are a set of	industry d	ummies def	ined at the	two-digit in	put-output
level, and a set of country dummes.										

Table 11	Credit market development interaction results, additional robustness checks
----------	---

Log GDP		Contract	Contract	Credit			
	P Log	enforcement	enforcement	collateral	Credit market		Entry
per capita	a population	cost	procedures	cost	development	Entry cost	procedures
log GDP per capita 1.000	00 -0.199	-0.538	-0.498	-0.365	0.642	-0.660	-0.504
og population	1.000	-0.245	0.025	-0.095	0.063	-0.071	0.196
Contract enforcement cost (% of debt)		1.000	0.120	0.210	-0.358	0.260	0.101
Contract enforcement procedures (/10)			1.000	0.212	-0.509	0.407	0.679
Credit collateral cost (% GDP per capita)				1.000	-0.344	0.417	0.306
Credit market development (% GDP)					1.000	-0.478	-0.387
Entry cost (% GDP per capita)						1.000	0.517
Entry procedures (/10)							1.000

Table A1	Correlation coefficients of country-level variables
----------	---

Table A2Estimated industry dummies for vertical integration

	Estimated		
	dummy		Coefficient
Industry (brief description)	coefficient	Industry (continued)	(continued)
Health/Education Services	-5.93	Radio/TV Broadcasting	-1.68
Maintenance Construction	-5.53	Manufacturing, Misc.	-1.56
Furniture, Household	-5.42	Machinery, Farm	-1.50
Household Appliances	-5.39	New Construction	-1.31
Automotive Service	-5.10	Machinery, Service Industry	-1.27
Wood Containers	-4.70	Industrial Equipment	-1.18
Eating/Drinking Places	-4.31	Utilities	-1.15
Furniture, Commercial	-4.27	Food	-0.91
Lodging/Personal Services	-4.23	Rubber	-0.84
Ordnance	-3.90	Paints	-0.79
Machinery, Industrial	-3.73	Textiles, Fabricated	-0.63
Ag/Forestry/Fishery Services	-3.72	Finance/Insurance	-0.49
Screw Machine/Stamping	-3.63	Mining, Chemical	0.05
Electrical, Misc.	-3.55	Engines	0.09
Footwear/Other Leather	-3.47	Motor Vehicles	0.38
Electric Lighting	-3.44	Real Estate	0.65
Scientific Instruments	-3.31	Transportation	1.26
Mining, Nonmetallic	-3.04	Metal Containers	1.44
Printing/Publishing	-3.04	Aircraft	1.53
Other Transportation Equipment	-2.89	Iron/Steel Manufacturing	1.57
Heating/Plumbing Fabrication	-2.87	Petroleum Refining	1.79
Optical Equipment	-2.81	Drugs/Cleansers	2.38
Machinery, Special	-2.68	Glass	2.66
Audio/Video Equipment	-2.65	Plastics	2.90
Machinery, Metalworking	-2.56	Computer/Office Equipment	3.31
Paperboard Containers	-2.52	Mining, Coal	4.71
Stone/Clay	-2.37	Electronic Components	4.91
Forestry/Fishery	-2.19	Nonferrous Metal Manufacturing	5.91
Lumber and Wood	-2.13	Apparel	6.19
Machinery, Mining	-2.13	Mining, Iron	6.50
Professional Services	-2.12	Communications, Not Radio/TV	6.94
Electrical Equipment	-2.03	Chemicals	9.02
Other Agricultural	-2.02	Fabrics	9.58
Other Fabricated Metal	-2.01	Amusement	10.06
Tobacco	-2.01	Livestock (omitted in regression)	10.88
Paper	-1.99	Leather	11.70
Textiles, Misc.	-1.96	Petroleum and Gas	13.33
Machinery, Misc.	-1.83	Mining, Nonferrous	17.08

The table reports estimated dummy variables in a firm-level regression of vertical integration on industry dummies using U.S. data.

	Ň	ertical inte	gration and	l alternativ	/e measure	s of specif	ic institutio	ons (firm le	vel)			
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
				Depe	ndent varid	able is the	vertical in	tegration i	ıdex			
Contract enforcement time	0.09 (0.04)	0.02 (0.03)										
Procedural complexity			0.010 (0.006)	0.003 (0.004)								
Disclosure index					-0.14 (0.05)	-0.01 (0.03)						
Interest rate spread							0.037 (0.018)	-0.023 (0.013)				
Entry time									0.007 (0.003)	0.000 (0.002)		
Antitrust									~	~	-0.28 (0.10)	0.01 (0.07)
Log number of employees	0.26	0.05	0.27	0.05	0.28	0.05	0.26	0.04	0.27	0.05	0.25	0.04
Industry dummies	ON No	Yes	oN	Yes	oN	Yes	ON No	Yes	oN	Yes	oN	Yes
R-squared	0.01	0.40	0.01	0.40	0.01	0.40	0.01	0.40	0.01	0.40	0.01	0.40
Number of Observations	661,747	661,747	661,580	661,580	661,693	661,693	639,677	545,928	661,747	661,747	655,929	655,929
The table presents coefficier	nt estimates	s from reg	ressions of	a firm-lev	vel vertical	integratio	n index or	n specific r	neasures o	f institutio	ns. Robus	t standard
errors, adjusted for clusterit Diankov et al (2003) Proce	ng within c dural comr	countries, dexity is a	are in pare n alternativ	entheses. ve measure	Industry d e of contra	ummies ar cting costs	e included from Diar	l where nc nkov et al (ted. Cont 2003) tha	rtact enford It has been	cement tin shown sig	te is from nificant in
other work (see Acemoglu a	nd Johnsor	1, 2003).	The disclos	sure index	measures	disclosure	of compar	iy ownersh	ip (see W	orld Bank,	2004). T	ne interest
rate spread is the spread bety	veen lendir	ig and bor	rowing rate	the cc	untry (fror	n year 200	0 World B	ank data).	Entry tim	le is from I	Jjankov et	al (2003).
I he antitrust measure is base	od on surve	y data, wit	th higher va	alues of th	e index rep	resenting	stronger ar	ititrust regi	ilation, as	in Dutz an	d Hayrı (1	.(664

Table A3